Author's response to reviews

Title: Identifying competencies required for medication prescribing for general practice residents: a nominal group technique study.

Authors:

Jean-Pascal Fournier (jean-pascal.fournier@univ-tlse3.fr)
Brigitte Escourrou (brigitte.escourrou@dumg-toulouse.fr)
Julie Dupouy (julie.dupouy@dumg-toulouse.fr)
Michel Bismuth (michel.bismuth@dumg-toulouse.fr)
Jordan Birebent (jordan.birebent@dumg-toulouse.fr)
Rachel Simmons (rachel.simmons@mail.mcgill.ca)
Jean-Christophe Poutrain (jc.poutrain@dumg-toulouse.fr)
Stéphane Oustric (stephane.oustric@dumg-toulouse.fr)

Version: 5 Date: 25 July 2014

Author's response to reviews: see over
July 25 2014

Dear editor,

Thank you for your consideration of our manuscript. Please find attached our fourth revision of the manuscript. We hope to have successfully answered the issues that you have raised.

Please do not hesitate to contact us if you have any further questions or comments regarding our manuscript.

We are looking forwards to your decision.

Sincerely yours,

Dr. Jean-Pascal Fournier
Département Universitaire de Médecine Générale
133 route de Narbonne, 31062 Toulouse
fax number: 335 5 62 88 90 84, telephone number: 335 62 88 90 48
email: jean-pascal.fournier@univ-tlse3.fr
1. Please remove the tracked changes from the additional files.
The tracked changes version has been removed from the additional files.

2. Please include the full name of the ethics committee that approved the study and a statement indicating whether written informed consent was obtained from the participants.

Accordingly, the methods have been implemented:

“Ethics approval for this study was obtained from the ‘Commission Ethique du Département de Médecine Générale de Midi-Pyrénées’ ethics committee. Participation in the study was entirely voluntary; there was no monetary reward for participation. Informed consent was obtained from all participants. The data remains anonymized and confidential.”